Value pluralism, trade-offs and efficiencies

PAPE **NGRKING**

by Nikolai Hoberg and Stefan Baumgärtner

University of Lüneburg Working Paper Series in Economics

No. 311

October 2014

www.leuphana.de/institute/ivwl/publikationen/working-papers.html

ISSN 1860 - 5508

Value pluralism, trade-offs and efficiencies

NIKOLAI HOBERG^{*a*} and STEFAN BAUMGÄRTNER^{*a,b,**}

^a Department of Sustainability Sciences and Department of Economics, Leuphana University of Lüneburg, Germany

^b Faculty of Environment and Natural Resources, University of Freiburg, Germany

October 1, 2014

Abstract: Political decision-makers face increasing demands to justify their actions with respect to multiple normative objectives or values. Here, we provide a general philosophical-economic clarification of the discussion of value pluralism, value-efficiency (i.e. efficiency with respect to values) and feasible relationships between values. We derive what relationships between values can exist in states of affairs that are valueefficient. In particular, we show that in a value-efficient state of affairs there is a tradeoff between at least two values, and that a win-win relationship between values is no indication of value-inefficiency when there are three or more values. Further, we connect these relationships to the opportunity costs of attaining a value to a higher degree. Lastly, we contrast Pareto-efficiency with value-efficient state of affairs. Our analysis provides insights that help decision-making in situations where there are multiple values at stake.

JEL-Classification: D63, H43, H00

Keywords: efficiency, value pluralism, value trade-offs, welfarism

^{*}Corresponding author: Chair of Environmental Economics and Resource Management, University of Freiburg, Tennenbacher Str. 4, D-79106 Freiburg, Germany, phone: +49.761.203-3753, fax: +761.203-3690, email: stefan.baumgaertner@ere.uni-freiburg.de.

1 Introduction

Political decision-makers face increasing demands to justify their actions with respect to multiple normative objectives or values. Potential candidates for fundamental values are, for example, equality, efficiency, freedom, security, or justice. Whatever the values, there are two central questions (e.g. Barry 1965): first, about the *feasibility* of combinations of degrees of attainment of values; second, about the *desirability* of combinations of degrees of attainment of values. For example, LeGrand (1990) asks these questions with regard to the trade-off between equality and efficiency: Does an increase in income equality reduce economic growth? How should society balance the objectives relating to income equality and economic growth? Here, some (e.g. Dasgupta 2005, Cowen 2007) argue that answering the first question is more pressing because merely clarifying the feasibility of actions can help decision-making and reduce conflict between different ethical perspectives. For clarifying feasibility, it is important to know the (factual) relationship between values, e.g. whether there is a trade-off or a win-win relationship between income equality and economic growth.

Apart from the large literature on the equality-efficiency trade-off (Okun 1975, Putterman et al. 1998, Blank 2002), there are many more results on relationships between other values in the economics literature, such as between income inequality and intergenerational income mobility (Björklund and Jäntti 1997, Corak 2013), between income inequality and subjective happiness (Alesina et al. 2004), between economic freedom and economic growth (DeHaan et al. 2006), between poverty reduction and environmental degradation (Baland et al. 2010), or between Pareto-efficiency and poverty reduction in payments for environmental services (Engel et al. 2008). The relationships between values also feature in political debates, for example, in the one on the social effects of economic growth such as the World Bank's report on green growth (World Bank 2012).

Our aims here are (i) to provide a general philosophical-economic clarification of the discussion of value pluralism and relationships between values; (ii) to contribute to the discussion by deriving a number of original statements; (iii) and, thus, to provide insights that help decision-making in situations where there are multiple values at stake. We proceed in two steps: Firstly, to conceptualize relationships between values carefully, we build on the philosophical literature for a definition of *value* in ethical theories and the debate on value pluralism. For example, it must be clear why equality and efficiency are values and why the relationship between the two is normatively meaningful. In this context it must also be clear whether equality and efficiency are instrumental to some other value such as social welfare or whether they are both intrinsic values. Secondly, we centrally employ the criterion of *value-efficiency*, i.e. efficiency with respect to values. This efficiency criterion is uncontroversial because it does not require weighing the degree of attainment of different values. Rather, it defines a state of affairs as value-efficient, if it is impossible to increase the degree of attainment of one value without necessarily reducing the degree of attainment of any other value.¹ This notion of efficiency has been more or less explicitly discussed in the philosophical and economic literature (e.g. Sen 1979a, LeGrand 1990, Raz 1997, Dasgupta 2005, Pattanaik and Xu 2012), but this literature has not directly discussed the question of how value-efficiency connects to relationships between values such as trade-offs and win-win relationships.

As a result, we derive what relationships between values can exist in states of affairs that are value-efficient. In particular, we show that in a value-efficient state of affairs there is a trade-off between at least two values, and that a win-win relationship between values is no indication of value-inefficiency when there are three or more values. Further, we connect these relationships to the opportunity costs of attaining a value to a higher degree. Lastly, we contrast Pareto-efficiency with value-efficiency and show that there can be a win-win relationship between values in a Pareto-efficient state of affairs.

This provides a unifying perspective for the interpretation of the many specific results on relationships between particular values such as the examples given above on relationships between economic growth, income equality and economic freedom. Also, these results on relationships between values and value-efficiency provide insights that help decision-making in situations where there are multiple values at stake.

The paper is structured as follows. Section 2 reviews the philosophical literature

¹It is undisputed that inefficient policy-making is considered bad. For good policy-making to be good it is necessary to be efficient.

and gives a definition of value. Section 3 introduces a microeconomic model, gives a definition of value-efficiency and a definition of different relationships between values. Section 4 discusses the connection between Pareto-efficiency and value-efficiency. Finally, Section 5 concludes.

2 Philosophical background

2.1 Definition of value

For the analysis of relationships between values one must be clear on what a *value* is and what it is not. Many definitions² of value focus on 'good' states of affairs. The examples from the introduction on values such as equality or efficiency also imply statements on good states of affairs, i.e. a state of affairs is better if it is more equal or more efficient. In light of this, we define a value in the spirit of Hurka (2006) and Chang (1997b) as follows:

Definition 1 (Value)

A *value* is a consideration which allows a comparison of states of affairs in terms of their goodness.

This definition refers to *states of affairs* which excludes rules and procedures from the definition of value.^{3,4} A *comparison* examines the differences between different states

²For Chang (1997b: 5) a value "is any consideration with respect to which a meaningful evaluative comparison can be made". Hurka (2006: 357) says that a theory of value determines "which states of affairs are intrinsically good and which intrinsically evil". Zimmerman (2010) refers to Scanlon (1998: 97) who has called the relationship between valueableness, goodness and intrinsic properties "a buck-passing account, since it "passes the buck" of explaining why something is worthy of being valued from its goodness to some property that underlies it" (Zimmerman 2010: Sec. 2).

 3 A "state of affairs" corresponds to what Chang (1997b), more generally, calls a "bearer of value" or "items".

⁴ Sen (2000) suggests a broad conception of states of affairs and refers to "comprehensive outcomes" that include the processes of choice such as actions performed and underlying motivations as well as final outcomes. We define states of affairs more narrowly in order to relate to the established literature.

of affairs in terms of a *consideration*. What constitutes the *goodness* of a state of affairs is determined in an ethical theory. For example, the conception of goodness in 'outcome utilitarianism' says that a state of affairs is better than another one if its sum of individual utilities is larger (Sen 1979b). Of course, there can be different conceptions of goodness, for example, there can be different equity criteria over individual utilities (Sen 1979a: 548) or non-preference metrics for individual advantage such as the capability approach (see e.g. Sinnott-Armstrong 2012).

The definition of value only allows to compare states of affairs in terms of their goodness. It does in itself not answer the fundamental question in ethics on how to act. One answer to this question is provided in 'act consequentialism' which says that an action is right if it brings about a state of affairs which is at least as good as each alternative state that results from any other feasible action (Sen 1979b: 464). That is, an action is right if it brings about the best consequences (Hurka 2006: 357). For example, 'act utilitarianism' says that an action is right if it leads to a state of affairs with a larger sum of individual utilities than any other feasible action. That is, act utilitarianism is based on act consequentialism and outcome utilitarianism (Sen 1979b).

Consequentialist ethical theories exclude other ethical theories such as deontological ethics which determines the right action without reference to good consequences or virtue ethics which relies on questions of moral character (Copp 2006: 20). This also means that relationships between values are most relevant for consequentialist ethical theories. For example, if there is a trade-off between values then determining the right action under consequentialism may require balancing of the different degrees of attainment of values with respect to a fundamental value. This raises the further question whether there is a 'fundamental' or intrinsic value. We follow Zimmerman (2010: Sec. 2) to define an intrinsic value as follows:

Definition 2 (Intrinsic value)

An *intrinsic value* is a value which allows a comparison of states of affairs in terms of their intrinsic goodness, i.e. goodness that is not derived from some other goodness.

For example, the intrinsic value in Bentham's utilitarianism (1907 [1789], 1988 [1776])

is aggregate happiness as it is not derived from some other value. A counterexample is an instrumental value that contributes to an intrinsic value.^{5,6} For example, Paretoefficiency is an instrumental value to social welfare as a Pareto-inefficient state can never maximize any function defined over individual utilities. Likewise, political liberties are sometimes seen as instrumental values to ensure other liberties such as civil liberties and liberty of conscience (e.g. in Rawls 1987: 13).

With this philosophical framework one can distinguish between an intrinsic value such as equality in egalitarianism and an instrumental value such as Pareto-efficiency for social welfare. This distinction is important because formulating relationships between instrumental and intrinsic values can be problematic, for instance, if not all intrinsic values are clearly defined. For example, some have questioned whether the trade-off between efficiency and equity is a normatively meaningful relationship because efficiency is an instrumental value and equity is an intrinsic value (LeGrand 1990). In this sense, a trade-off between equity and efficiency begs the question what the intrinsic value behind efficiency is. If the trade-off is assumed to be between intrinsic values, then this requires justification why Pareto-efficiency is an intrinsic value in the same way as equity (LeGrand 1990: 566).

2.2 Value pluralism

With regard to intrinsic values, there is the question if there is one single intrinsic value to which all values can be reduced to. The idea that all values can be reduced to one single intrinsic value is called *value monism* (Mason 2011). An example of value monism is Bentham's utilitarianism (1907 [1789], 1988 [1776]) where only aggregate happiness

 $^{^{5}}$ On the question whether every value that is not intrinsic is an instrumental value see Zimmerman (2010: Sec. 2)

⁶ Chang (1997b: 9) uses a similar framework when she talks about values. For her bearers of value (here state of affairs) are compared in terms of their merits (here attributes) with respect to a covering value (here intrinsic value). And a covering value (here intrinsic value) can rely on different contributory values (here instrumental values). She then defines comparativism as the view that justified choice requires the comparison of bearers of values.

is intrinsically good and all other values are instrumental to this one intrinsic value. Under value monism one can formulate relationships between instrumental values. For example, if social welfare is the intrinsic value this allows trade-offs between different constituents of well-being, such as consumption, leisure or health and education (e.g. in Dasgupta 2005: 241).

If it is impossible to reduce all values to one intrinsic value, this is value pluralism:

Definition 3 (Value Pluralism)

An ethical theory is characterized by *value pluralism* if it is based on multiple intrinsic values.⁷

Value pluralism is found in the work of philosophers such as Isaiah Berlin (Berlin 1969) who speaks of different concepts of liberty and other values such as equality or justice. Also, Brian Barry speaks of a "plurality of 'ultimate' values" (Barry 1965: 5) such as equality and freedom when he discusses political arguments on normative objectives. Ethical theories under value pluralism allow the formulation of relationships between intrinsic values. For example, under value pluralism there can be a trade-off between the intrinsic value of social welfare and the intrinsic value of nature. Similarly, one could think of the relationship between intra- and intergenerational justice in sustainability problems as a relationship between intrinsic values (e.g. Baumgärtner et al. 2012).

With regard to desirability, value pluralism can create problems for decision-making such as incomparability which results from incomplete orderings of a set of states of affairs (see Sen 1985: 179, Chang 1997b).⁸ For example, it can be that there are two acts that lead to two states of affairs where one has a higher degree of attainment of

⁷ Sen (1985: 178) argues that value pluralism is different from informational pluralism: That is, there can be one intrinsic value (value monism) which subsumes many different kinds of information such as one conception of social welfare that relies on utility and non-preference information. Or there can be multiple intrinsic values (value pluralism) that are all based on the same kind of information such as different social welfare functions that are all based on utility-information.

⁸ Hsieh takes 'incommensurability' to refer to the relation between values in the abstract sense and 'incomparability' to refer to the relation between concrete bearers of value which are states of affairs in this analysis (Hsieh 2008: Sec. 1.2).

one intrinsic value such as intergenerational equity, and the other has a higher degree of attainment of another intrinsic value such as intragenerational justice. In order to determine one right action in this context, one cannot use another more 'fundamental' value to balance the degree of attainment of the two values as both are intrinsic values. Nevertheless, there exist several approaches to decision-making under value pluralism (e.g. Mason 2011: Sec. 4).⁹

Value pluralism and value monism comprise different ethical theories which has consequences on how the set of feasible states of affairs in terms of values is defined. Ethical theories under value monism view the set of feasible states of affairs in terms of one intrinsic value and multiple instrumental values. This means there can be trade-offs between instrumental values such as equality and economic growth with regard to an intrinsic value such as utilitarian social welfare. Ethical theories under value pluralism view the set of feasible states of affairs in terms of multiple intrinsic value and multiple instrumental values. This means there can be trade-offs between intrinsic values such as social welfare and the intrinsic value of nature. Also, there can be normatively meaningful trade-offs between instrumental and intrinsic values under value pluralism. Yet, these require that all intrinsic values are explicated to allow a clear interpretation of this relationship.

The concrete form of the set of states of affairs plays a big role in the difficulty of decision-making, irrespective of the distinction between value pluralism and value monism. For example, if the feasible set does not force one into making trade-offs be-

⁹ Mason (2011: Sec. 4) lists the following approaches: The first approach is practical wisdom (e.g. Anderson 1993, Nagel 1979). Practical wisdom solves problems of incomparability due to multiple intrinsic values without reasoning from general principles, but rather with a faculty of judgement. The second approach uses covering values to determine the weight of each intrinsic value for each circumstance of decision-making (e.g. Chang 1997b). These covering values are said not to be a case of value monism as they always depend on the concrete choice situation. The third approach (suggested by Joseph Raz 1999) who says that one is free to choose among "rationally eligible" options, i.e. states of affairs that are not dominated by others. The fourth approach, says that rather than trying to achieve a rational choice under value pluralism one should accept irresolvable conflicts between values (Williams 1981).

tween values, decision-making is easy – the higher attainment of one value necessarily increases the attainment of any other value and one can attain a value without compromising any other value. For example, if the feasible set showed that equality and economic growth are always in a win-win relationship, this would also ease political debates on this issue considerably. On the other hand, if the feasible set forces one constantly into making trade-offs between values, then decision-making is rather more difficult. In the following, we therefore want to focus our analysis on the feasible set and the relationships between values.

3 Formal model and analysis

3.1 Actions and states of affairs

In order to relate the model to the discussion on ethics and act consequentialism in particular, we consider states of affairs, which can be thought of as the outcome of a particular action. A state of affairs $y \in Y \subseteq \mathbb{R}^m$ is a complete description of the world in terms of m different attributes, so that $y = (y_1, ..., y_j, ..., y_m)$. Thus, the set Y can be thought of as the set of feasible states of affairs in terms of attributes. These attributes y_j contain continuously measurable information on, for example, stocks such as physical, social, human and natural capital; on flows such as payoffs, income, and environmental (dis-)services; and on individuals such as their number, the distribution of income, capabilities, and resources over individuals. We assume that the set of states of affairs Y is non-empty, closed and bounded and, thus, a compact set. Further, superscripts denote different states of affairs, such as y^a and y^b . Here, the assumption is that all these attributes are continuously measurable or at least representable, prima facie, via indicators, which allow to measure distance in Y-space.¹⁰ For example, a state of affairs could contain attributes on the income distribution or the level of gross domestic product (GDP).

 $^{^{10}}$ This excludes the use of discrete attributes as in e.g. Pattanaik and Xu (2012).

3.2 Value relation and single-value index function

As discussed in Section 2 different ethical theories determine values differently. Therefore, the value set V contains l = #V values where $v \in V$ denotes a value in the set V. The concrete nature of these values and the number l of values are determined within a given ethical theory. In this way, the value set should only contain values which allow meaningful relationships (see Section 2). For example, under value monism a value set V with l = 1 contains one intrinsic value such as utilitarian social welfare. In the same way, V could contain multiple values l > 1 under value monism if there are different instrumental values to the single intrinsic value. Alternatively, under value pluralism, Vwith l > 1 could contain different intrinsic values. As we are interested in relationships between values, formulations of value sets with more than one value are most relevant to our analysis.

For the comparison of states of affairs there exists a value relation for each value.

Definition 4 (Value relation)

A value relation \succeq_v is a binary relation on the set of states of affairs Y with respect to a value $v \in V$, where $y^a \succeq_v y^b$ means that y^a is as at least as good as y^b in terms of value v.

For example, under egalitarianism income inequality could provide a value relation for the value of equality. In economics the properties of value relations are discussed in the social choice and fair social orderings literature (e.g. Fleurbaey et al. 2005). Also, in philosophy there has been some discussion on the properties of value relations in recent years (e.g. Rabinowicz 2008). The traditional framework includes three value relations: *better than, worse than* and *equally good as* ("trichotomy thesis") (Chang 1997b: 4).¹¹

¹¹These value relations can be derived from the relation \succeq_v as follows: better than: $y^a \succ_v y^b$ if and only if $y^a \succeq_v y^b$ and not $y^b \succeq_v y^a$; worse than: $y^a \prec_v y^b$ if and only if $y^b \succeq_v y^a$ and not $y^a \succeq_v y^b$; equal to: $y^a \sim_v y^b$ if and only if $y^a \succeq_v y^b$ and $y^b \succeq_v y^a$. Chang (1997a, 2002a, 2005) has suggested a fourth value relation that she called "on a par" where two states of affairs are comparable, yet not better than, worse than or equal to another. Boot (2009) comments that this debate is not directly relevant to justified choice as it creates the same problems as incomparability between bearers of value

The properties of value relations are directly relevant for decision-making under act consequentialism as these determine the ordering of states of affairs with respect to a value (see e.g. Sen 1970b: 9). Common assumptions on the properties of value relations are the following:

Definition 5

A value relation \succeq_v is

- reflexive, if and only if for all $y^a \in Y$ one has: $y^a \succeq_v y^a$
- transitive, if and only if for all $y^a, y^b, y^c \in Y$ one has: $y^a \succeq_v y^b$ and $y^b \succeq_v y^c$ implies $y^a \succeq_v y^c$
- complete, if and only if for all $y^a, y^b \in Y$ one has: $y^a \succeq_v y^b$ or $y^b \succeq_v y^a$ (or both)
- anti-symmetric: if and only if for all $y^a, y^b \in Y$ one has: $y^a \succeq_v y^b$ and $y^b \succeq_v y^a$ implies $y^a = y^b$
- continuous, if and only if for any sequence of pairs $\{(y_n^a, y_n^b)\}_{n=1}^{\infty}$ with $y_n^a \succeq_v y_n^b$ for all n, $y^a = \lim_{n \to \infty} y_n^a$, and $y^b = \lim_{n \to \infty} y_n^b$, one has $y^a \succeq y^b$
- *lexicographic* for two attributes¹², if and only if there exist two attributes $y_i, y_j \in \{y_1, ..., y_m\}$ for which one has: $y^a \succeq_v y^b$ if and only if $y_i^a > y_i^b$ or $y_i^a = y_i^b$ and $y_j^a \ge y_j^b$.

Each of these assumptions on value relations can be, and has been, criticized in philosophy. For example, Broome (1991: 18) considers reflexivity and transitivity as the prime characteristics of rationality. Hausman (1993) criticizes just these characteristics with respect to value relations due to the vagueness of some normative predicates. The assumption of completeness excludes cases where no value relation holds between states

where no value relation holds. Thus, for the course of this analysis we will not consider parity as a fourth value relation.

¹²This is a very special lexicographic ordering of attributes but which still captures the intuition on priority of an increase in one attribute over any increase in other attributes. A more general treatment of lexicographic relations can be found in e.g. Hougaard and Tvede (2001).

of affairs and, therefore, excludes problems of incomparability and incommensurability which are sometimes highlighted in philosophical reasoning (e.g. Chang 1997a, Hsieh 2008). Still, completeness of each respective value relation allows us to model incomparability. That is, two states of affairs $y^a, y^b \in Y$ can be incomparable in terms of values, if two intrinsic values $v, w \in V$ individually can provide a complete ordering of states of affairs, but differ in their ordering regarding of the two states, e.g. $y^a \succ_v y^b$ and $y^a \prec_w y^b$. Then, both states of affairs are ordered completely with respect to $v, w \in V$, but y^a is better than y^b in terms of v and worse in terms of w, so that the two states of affairs are incomparable in the sense that it is impossible to rank one over the other.

A prominent discussion on a complete and transitive value relation is the social ordering in Arrow's impossibility theorem (Arrow 1951). It shows the impossibility of establishing a complete and transitive social choice function from ordinal non-comparable utility information and further axioms (see e.g. Roemer 1996). Still, there are different interpretations of Arrow's impossibility theorem. A rather optimistic one is provided by Sen (1999) who highlights the different possibilities of making interpersonal comparisons in terms of incomes, primary goods, or basic needs. The difficulty in finding and constructing such complete and transitive social orderings for the social evaluation of states of affairs has long been debated for example in the related discussion on indices for distributive justice (e.g. Fleurbaey 2007). A salient example is Rawls's (1971) theory of justice which uses primary goods as its metric of individual advantage to identify the worst-off individuals which creates the need to construct a suitable ordering for the comparison of individuals.

The more technical assumption of continuity excludes some value relations that might be relevant in ethical theories. For example, lexicographic value relations are not continuous as they give priority to an increase in one attribute irrespective of any increases in some another attribute. An example for this is John Ralws lexicographic ordering of basic liberties before prosperity in his difference principle (Rawls 1971: Sec. 8) or James Griffin's (1986: 83) discussion of cases where one attribute trumps another.

Different combinations of these assumptions establish different orderings over the set of states of affairs (as in Sen 1970b: 9). In this way, these assumptions are linked to a maximizing account of consequentialism which says an action is right if it leads to a state of affairs with the highest possible degree of attainment of a value (e.g. Broome 1991, Sinnott-Armstrong 2012).

Under certain assumptions on value relations there exists an index function that represents the ordering of a value relation.

Definition 6 (Index function)

A function $I_v: Y \to \mathbb{R}$ is an *index function* representing value relation \succeq_v if and only if, for any $y^a, y^b \in Y$,

$$y^a \succeq_v y^b \Leftrightarrow I_v(y^a) \ge I_v(y^b)$$

The index function I_v is an index of value v and shows the degree of attainment of this value.

Lemma 1 (Existence of index function)

If the value relation \succeq_v is complete, transitive and continuous on Y, then there exists a continuous index function $I_v: Y \to R$ which represents the value relation \succeq_v

Proof. By analogy the standard proof for the existence of a utility function applies, see e.g. MasColell et al. (1995: 47). \Box

An example of such an index function is a utility function which show the degree of attainment of the value of an individual preference. Another example are Bergson-Samuelson social welfare functions $W(\cdot)$ which show the degree of attainment of social welfare as a function of individual utilities U_i (e.g. Suzumura 1987). Despite Arrow's impossibility theorem and the associated debate on its axiomatic foundation, social welfare functions that are based on interpersonally comparable utility information are used in areas of applied economics such as the economics of climate change (e.g. Stern 2007).

3.3 Value pluralism and value-efficiency

As discussed in Sections 2 and 3.1, the value set V can contain one single or multiple values depending on the respective ethical theory. For example, it can contain one intrinsic value under value monism or multiple intrinsic values under value pluralism. Given such a value set V one can define value-efficiency as follows:

Definition 7 (Value-efficiency)

A state of affairs $y \in Y$ is *value-efficient* if and only if there exists no other state of affairs $y' \in Y$ for which $y' \succeq_v y$ for all $v \in V$ and $y' \succ_v y$ for at least one $v \in V$.

This criterion is in line with the maximizing account of consequentialism, as discussed above, where a state of affairs with a higher attainment of one value is always better than one with a lower degree of attainment. Indeed, value-efficiency provides some guidance for actions as it characterizes those actions as bad that lead to value-inefficient states of affairs. As value-efficiency itself is not a very demanding concept it provides room for many consequentialist ethical theories that are based on different definitions of value. Some authors have gone further and suggested value-efficiency as a criterion for choosing states of affairs under value pluralism. For example, Raz (1997) says that under intrinsic value pluralism any value-efficient state of affairs can be chosen with free will (what he calls 'rationally eligibile' states of affairs). Similarly, Brun and Hirsch-Hadorn (2008) determine value-efficient states of affairs concerning the multiple values within the sustainability concept.

In economics this notion of value-efficiency has also been mentioned and discussed. For example, Sen (1979a: 553) refers to 'dominance' with respect to normative considerations and discusses its difference to Pareto-efficiency under value pluralism. LeGrand (1990: 559) specifically uses the notion of value-efficiency in his discussion of the equityefficiency trade-off. Pattanaik and Xu (2012) discuss dominance with respect to evaluative attributes and the use of context-specific information in decision-making. Others have noted the limited usefulness of value-efficiency for decision-making as it provides only incomplete orderings of states of affairs (Sen 1985: 178). Finally, Dasgupta (2005) shows the range of values that can be incorporated in economic analysis by distinguishing Pareto-efficiency which is based solely on utility information from broader 'efficiency' which is based on utility and non-preference information. Based on these specific contributions, we will further discuss the relationship between Pareto-efficiency and value-efficiency in Section 4.

With Definition 7 of value-efficiency, one may ask under what conditions there exist value-efficient states of affairs for different values. If the set of states of affairs Y is a finite set and \succeq_v satisfies reflexivity, completeness and transitivity for all $v \in V$, then there exists at least one value-efficient element (Sen 1970b: 30). If Y is an infinite set and \succeq_v additionally satisfies continuity, then there exists at least one value-efficient element due to the Bolzano-Weierstrass theorem. There exists a broad literature on the existence of maximal elements with weaker continuity and transitivity axioms following Bergstrom (1975) and Walker (1977). For the case of lexicographic value relations there are several results on the existence of value-efficient states (e.g. Houy and Tadenuma 2009). Altogether, the discussion shows that there exist value-efficient state of affairs for values that satisfy many different properties and for different sets of states of affairs. For example, there exists a value-efficient state if some values are lexicographic, such as priority of the worst-off, and others are continuous, such as indices for income inequality.

3.4 Relationships between values

As argued in Section 2 the form of the set of feasible states of affairs and relationships between values impact on the difficulty of decision-making. This leads us to distinguish the following relationships¹³ between two values. Also, for value sets with more than two values l > 2, we will continue with pairwise relationships between two values.

Definition 8 (Relationships between values)

In a feasible state of affairs y a *relationship* R between two values $v, w \in V$ is a binary relation on $V \times V$, denoted as vRw, with the following special cases:¹⁴

(i) R is a trade-off relationship if and only if $y' \succ_v y$ implies $y' \prec_w y$ for all $y' \in Y$

¹³ These relationships are common in many discussions, e.g. LeGrand (1990), Engel et al. (2008), Glotzbach and Baumgärtner (2012), and Baumgärtner et al. (2012).

¹⁴More generally, relationships may be defined for a local environment around the state of affairs $y \in Y$ in which the condition(s) must hold. This leads to the additional condition on y' that $||y'-y|| \leq \varepsilon$ for $\varepsilon > 0$.

- (ii) R is a win-win relationship if and only if $y' \succ_v y$ implies $y' \succ_w y$ for all $y' \in Y$
- (iii) R is an *independence* relationship if and only if there exists $y', y'' \in Y$ such that $y' \succ_v y$ implies $y' \succ_w y$ and $y'' \succ_v y$ implies $y'' \sim_w y$

The relationships R are defined as directed relationships from one value v on another value w. The converse relationship only holds if R is symmetric.^{15,16}

The most familiar relationship (i) is the one of a *trade-off* between values. A tradeoff is the case when attaining one value to a higher degree necessarily reduces the degree to which one attains the other one. This is a symmetric relationship which holds both ways. An example from the introduction is the trade-off between efficiency and equality.

The next relationship (ii) is the one of a *win-win relationship* between values. A winwin relationship is the case where achieving one value facilitates achieving the other one, that is, attaining one value to a higher degree induces a higher degree of attainment of the other one. This is an asymmetric relationship which does not necessarily hold both ways. An example from the introduction is the win-win relationship between income equality and intergenerational income mobility.

The last relationship (iii) is one of *independence* between values. Independence is the case when values can be achieved independently, that is, attaining one value to a higher degree does not necessitate any change in the degree to which one attains the other one. An example is the case where the intragenerational distribution of CO_2 emission permits is independent from the intergenerational distribution of CO_2 emission permits, so that the attainment of intragenerational distributive justice is independent of the attainment of the attainment of intergenerational distributive justice.

These relationships are illustrated in Figure 1 for the case of a feasible set of states

¹⁵A trade-off relationship is always symmetric in that one value trades-off with another. Win-win relationships and independence relationships are not symmetric. For example, it can be that relationships are only onesided as illustrated below in Figure 1.

¹⁶If some value relation does not satisfy completeness, then there could be incomparability as a relationship where one value is attained to a higher degree $y' \succ_v y$, the two states are incomparable with regard to another value w.

of affairs (depicted in value-space) with two values $v, w \in V$.¹⁷ The axes in this figure show the degree of attainment of values v and w via the respective index functions I_v and I_w . A point in this figure represents a state of affairs such as y^b which results from an action. The black boundary line is the frontier which delimits states of affairs that are feasible, those inside the grey area, from states of affairs that are not feasible, those inside the white area. The north-eastern part of the frontier, between y^c and y^d shows all value-efficient states of affairs. In y^a there is an independence relationship between



Figure 1: Convex set of feasible states of affairs in value-space for two values $v, w \in V$. The axes show the degree of attainment of values v and w via the respective index functions I_v and I_w . Each point in the figure corresponds to a state of affairs that results from a specific action. The black frontier delimits feasible states of affairs in the grey area from not feasible ones in the white area. Value-efficient states of affairs are on the frontier between y^c and y^d . There are different relationships in different states of affairs: Trade-off relationship in y^c, y^b, y^d ; Win-win relationship in y^e, y^f ; Independence relationship in y^a . Adapted from Baumgärtner et al. (2012).

¹⁷ The shape of the set of feasible states of affairs Y in value-space as in Figures 1 and 2 relies on further assumptions regarding the feasible set Y in attribute-space that go beyond the ones made in Section 3.

values v and w as the attainment of either value can be increased without necessarily changing the attainment of the other. In y^e there is a win-win relationship between values v and w as an increased attainment of w necessarily increases the attainment of v. Yet, this win-win relationship is only onesided as the attainment of v can be increased without necessarily increasing the attainment of w. In y^f the opposite case of win-win relationship is depicted. In any state of affairs along the frontier between y^c and y^d there is a trade-off relationship between the attainment of values v and w as attaining either one to a higher degree necessarily reduces the degree of attainment of the other one.

Figure 2 illustrates the relationships between values for the case of a feasible set of states of affairs (again depicted in value-space) with three values $u, v, w \in V$. Correspondingly, there are three axes in this figure which show the degree of attainment of values u, v and w via the respective index functions I_u , I_v and I_w . The surface of the set in value-space, which is indicated by the thin black lines, delimits states of affairs that are feasible, those inside the grey volume, from states of affairs that are not feasible, those inside the white volume. Also, the surface shows all value-efficient states of affairs. In y^a there is a trade-off relationship between v and w as increasing either one necessarily reduces the attainment of the other value. Also, there is an independence relationship between v and u as increasing the degree of attainment of v can either lead to an increased attainment of u (by going from y^a to y^c) or remain on the same level of attainment (by going from y^a to y^b). There are no win-win relationships in Figure 2 as in no state of affairs the increased attainment of one value necessarily increases the attainment of any other.

From Figures 1 and 2 the connection between a trade-off between values and efficiency becomes apparent:

Proposition 1 (Value-efficiency and relationships between values)

(i) If there exist at least two value-efficient state of affairs, then in every value-efficient state of affairs there exists at least one trade-off relationship between two values v, w ∈ V.



Figure 2: Convex feasible set in value-space for three values $u, v, w \in V$. The three axes show the degree of attainment of values u, v and w via the respective index functions I_u, I_v and I_w . Each point in the figure corresponds to a state of affairs that results from a specific action. The surface of the set in value-space, indicated by the thin black lines, delimits states of affairs that are feasible in the grey volume from states of affairs that are not feasible in the white volume. The surface also contains all value-efficient states of affairs. There are different relationships in between different values in a state of affairs: In y^a there is a trade-off relationship between v and w and independence relationship between u and v.

- (ii) If in a state of affairs $y \in Y$ there is a trade-off between at least two values $v, w \in V$, then y is value-efficient.¹⁸
- (iii) If there exist at least three values in the value set, $\#V = l \ge 3$, there can be winwin relationships or independence relationships between values in a value-efficient state of affairs.

Proof. See Appendix A.1

¹⁸If one considers locally defined relationships this may not hold as it may be that in a value-inefficient state there are trade-offs in a local environment.

This means that win-win relationships between two values do not necessarily indicate value-inefficient states of affairs. Rather, the absence of value trade-offs is a clear indication of a value-inefficient state of affairs. For example, suppose there are three values such as income equality, intergenerational income mobility and overall economic well-being as indexed by GDP. There can be a win-win relationship between income equality and intergenerational income mobility in a value-efficient state of affairs. Yet, it must be that there is a trade-off with GDP and economic well-being as this state of affairs would otherwise not be value-efficient. If the value set contains only two values #V = l = 2 then there is always a trade-off between these values in any value-efficient state of affairs. This can be illustrated in Figure 1 where the increased attainment of one value reduces the attainment of the other in all value-efficient states on the frontier between y^c and y^{d} .¹⁹

3.5 Opportunity costs

The concept of opportunity costs is closely related to different relationships between values. We define opportunity costs between values as follows:

Definition 9 (Opportunity costs)

The opportunity cost of value $v \in V$, in units of value $w \in V$, is the minimal amount, given the set of feasible states of affairs Y, of value w that one has to forego to attain an additional unit of value $v \in V$. In particular, if an index function exists for values $v, w \in V$, the marginal opportunity costs of achieving a value v to a higher degree, in units of value w foregone, in a value-efficient state $y \in Y$ are:

$$C_{v,w}(y) = dI_w(y)/dI_v(y)$$
 . (1)

¹⁹More generally, the analysis of pairwise relationships between two values proved limited for cases where more than two values are involved. For example, in this case the information on win-win or tradeoff relationships between two values must be interpreted much more carefully in light of the effects on other potentially involved values. A possibility would be to define binary relationships between values with the *ceteris paribus*-provision that *the degree of attainment of all other values remains constant*. This would ensure that there are only trade-offs between values in a value-efficient state of affairs.

Proposition 2 (Opportunity costs and relationships)

The sign of opportunity costs, if they are defined, corresponds to the relationship vRwbetween values $v, w \in V$ in a state of affairs $y \in Y$:

- (i) if R is a trade-off, then opportunity costs of v in units of w are positive.
- (ii) if R is a win-win relationship, then opportunity costs of v in units of w are negative.
- (iii) if R is an independence relationship, then opportunity costs of v in units of w are zero or negative.

Proof. See Appendix A.2.

This can be illustrated in Figure 1 for the case of two values $v, w \in V$. In any state of affairs along the frontier between y^c and y^d there are positive opportunity costs of attaining either v or w to a higher degree as there is a trade-off relationship in these states. In y^a there are negative or zero opportunity costs to increasing either v or w to a higher degree as there is an independence relationship in y^a . In y^e there are negative opportunity costs to increasing w in units of v as there is a one-sided win-win relationship in y^e . The opposite case is in y^f where there are negative opportunity costs to increasing v in units of w as there is the reverse one-sided win-win relationship in y^f .

In Figure 2 opportunity costs can be illustrated for the case of three values $u, v, w \in V$. In y^a there are positive opportunity costs between v and w as there is a trade-off relationship between the two values in y^a . Also, there are negative or zero opportunity costs between v and u as there is an independence relationship between these two values in y^a , depending if one moves from y^a to y^c or from y^a to y^b . There is no state in Figure 2 where there are only negative opportunity costs as there exists no state of affairs where there is a win-win relationship between any two values.

Corollary 1 (Opportunity costs and value-efficiency)

In a value-efficient state of affairs $y \in Y$ opportunity costs are as follows:

(i) If there exist at least two value-efficient state of affairs, then in every value-efficient state of affairs there exist positive opportunity costs between two values $v, w \in V$.

- (ii) If in a state of affairs $y \in Y$ there exist positive opportunity costs between at least two values $v, w \in V$, then y is value-efficient.²⁰
- (iii) If there exist at least three values in the value set, $\#V = l \ge 3$, there can be positive or zero or negative opportunity costs between values in a value-efficient state of affairs.

Proof. See Appendix A.3.

As illustrated in Figure 1 and 2 there exist positive opportunity costs in value-efficient states for both the case of two values or three values in the value set. For the case of three values, it becomes clear that negative opportunity costs are not an indication of an value-inefficient state of affairs.

4 Value-efficiency and Pareto-efficiency

4.1 Individualistic framework

As the definition of value-efficiency and relationships between values is based on a very broad consequentialist definition of value, it is interesting to look at its connection to the familiar definition of Pareto-efficiency. This requires to introduce the familiar individualistic framework from welfare economics. In a first step, define a preference relation as special case of a value relation with respect to the value of an individual preference. Suppose there are z individuals whose self-regarding preferences p_k (with $k \in \{1, ..., z\}$) are contained in a set P with $\{p_1, ..., p_k, ..., p_z\} = P$.

Definition 10 (Preference relation)

For all individuals k = 1, ..., z, a preference relation \succeq_{p_k} of individual k is a binary relation on the set of states of affairs Y, where $y^a \succeq_{p_k} y^b$ means that y^a is as at least as good as y^b in terms of individual preference p_k .²¹

²⁰If one considers locally defined relationships this may not hold as it may be that in a value-inefficient state there are trade-offs in a local environment.

 $^{^{21}}$ The assumption that individuals have complete preferences over the whole set of state of affairs

In this setting, Pareto-efficiency can be defined as follows:

Definition 11 (Pareto-efficiency)

A state of affairs $y \in Y$ is *Pareto-efficient* if and only if there exists no other state of affairs $y' \in Y$ for which $y' \succeq_{p_k} y$ for all individual preferences $p_k \in P$ and $y' \succ_{p_k} y$ for at least one individual preference $p_k \in P$.

This definition of efficiency shows the liberal heritage of Pareto-efficiency which is based on the set P of individual preferences and the idea that all individuals should be free to pursue the satisfaction of their individual preferences.

4.2 Value sets and individual preferences

In the next step, we consider four different cases for the role of preferences in value sets to discuss the connection between value-efficiency and Pareto-efficiency.

The first case is the one where the value set contains all individual preferences and only individual preferences P = V in which case value-efficiency in Definition 7 reduces to Pareto-efficiency in Definition 11.

Proposition 3 (Equivalence of value-efficiency and Pareto-efficiency) Value-efficiency is equivalent to Pareto-efficiency if and only if P = V.

Proof. See Appendix A.4

The equivalence of value-efficiency and Pareto-efficiency is briefly discussed, for example, in Dasgupta (2005: 240) who discusses the case when all non-preference values are incorporated into a notion of individual utility. This case is well studied in welfare economics where Pareto-efficiency serves as a central normative criterion, especially following Arrow's impossibility theorem (Sen 1999: 352). Several authors have defended

Y where each state of affairs $y \in Y$ has m different attributes is stronger than the usual assumption that individuals only have preferences over their individual consumption sets. While it is possible to define preference relations over only particular attributes of a state of affairs that concern an individuals self-regarding preference, we do not do so here in order to keep formal requirements to a reasonable level that sill allows us to support our results.

this prominent use of Pareto-efficiency. For example, Buchanan says that "[s]ince 'social' values do not exist apart from individual values in a free society, consensus or unanimity (mutuality of gain) is the only test which can insure that a change is beneficial" (Buchanan 1959: 137). This exclusive reliance on individual preferences in the assessment of states of affairs leads to the debate on the ethical appeal of welfarism (e.g. Sen 1979b, Sen 1980, Roemer 1996, Kaplow and Shavell 2001, Fleurbaey et al. 2003). For example, this concerns the question if all values can be reduced to individual preferences and how values that deviate from individual preferences can be justified. Further, if the value set contains only individual preferences, the earlier results in Section 3.4 can be interpreted accordingly. For example, Proposition 1 says that there can be a win-win relationships between two individual preferences in Pareto-efficient states of affairs when there are trade-offs between other individual preferences.

The second case where only some individual preferences are included in the value set, $V \subset P$, is discriminatory. This would require an ethical theory where preferences of some people do not matter in the assessment of states of affairs. This of course includes cases where the value set does not contain all individual preferences and further nonpreference values.

There may be values that are not derived from individual preferences, sometimes called non-preference (or non-utility) values.

Definition 12 (Non-preference value)

A value v is a non-preference value if $v \in V \setminus P$.

For example, Dasgupta (2005: 240) mentions democracy and civil liberties as examples of such additional non-preference values. Another salient example, he mentions, is the liberal paradox from microeconomic theory where under certain conditions liberalism (as an inviolable personal domain) is in conflict with Pareto-efficiency (Sen 1970a).

Non-preference values allow the third case of value sets which are based ethical theories, especially the ones labeled 'non-welfarist', which do not base their value set V on the satisfaction of individual preferences, $P \cap V = \emptyset$. For example, Kaplow (2007) discusses the capability approach by Sen (1980) and primary goods in Rawls (1971) as

theories do not include individual preferences but rather the means to the fulfillment of individual preferences in their assessment of states of affairs. He goes on to discuss the conflict of these alternative ethical theories with Pareto-efficiency. Similarly, there is a discussion on the conflict of efficiency criteria for value sets derived from different ethical theories (e.g. Brun and Tungodden 2004, Fleurbaey 2007). This concerns the conflict between an efficiency criterion for one value set V (value-efficiency) which contains only non-preference values such as primary goods and another efficiency criterion on a set P(Pareto-efficiency) which contains only individual preferences.

A related case is the fourth case where the value set contains individual preferences and additionally non-preference values, $P \subset V$. In this case Pareto-efficiency and value-efficiency are not equivalent. Still, the inclusion of non-preference values into an economic framework can lead to a conflict with Pareto-efficiency in some sets of feasible states of affairs (e.g. Kaplow and Shavell 2001, Fleurbaey et al. 2003).

This last case can be illustrated in Figure 3. It shows a convex set of feasible states of affairs in value-space with three values: two individual preferences $p_1, p_2 \in P \subset V$ and one non-preference value $v \in V$. The axes in this figure show the degree of attainment of the three values via their respective index functions: two utility functions U_1, U_2 and an index function for a non-preference value I_v . The surface of the set in value-space, which is indicated by the thin black lines, delimits states of affairs that are feasible, those inside the grey volume, from states of affairs that are not feasible, those inside the white volume. Also, the surface shows all value-efficient states of affairs. All Pareto-efficient states of affairs are those on the dashed line on the surface where the non-preference value is attained to the lowest degree. Thus, surface of the set in value-space encompasses all value-efficient and all Pareto-efficient states of affairs.

Here, one can distinguish a Pareto-efficient and value-efficient state of affairs such as y^a from a value-efficient state such as y^b . States of affairs that are not on the surface are value-inefficient and Pareto-inefficient. In the following we focus on this case of preferences and non-preference values $P \subset V$ as this has attracted the most interest in economics.



Figure 3: Convex set of feasible states of affairs in value-space with three values. The axes in this figure show the degree of attainment of the three values via their respective index functions: two utility functions U_1 , U_2 and an index function for a non-preference value I_v . The surface of the set in value-space, which is indicated by the thin black lines, delimits states of affairs that are feasible, those inside the grey volume, from states of affairs that are not feasible, those inside the white volume. All value-efficient states of affairs are on the surface of the set in value-space. All Pareto-efficient states of affairs are those on the dashed line on the surface where the non-preference value is attained to the lowest degree. State of affairs y^a is Pareto-efficient and value-efficient, state of affairs y^b is value-efficient and not Pareto-efficient.

Corollary 2 (to Proposition 1 (iii))

If there are at least two non-preference values $v, w \in V \setminus P$ and at least two individual preferences $p_1, p_2 \in P \subset V$, then there can be any relationship between the two values v, w in a Pareto-efficient state of affairs $y \in Y$.

Proof. See Appendix A.5

This result is relevant to the interpretation of Proposition 1 as this means that there can be a win-win relationship between non-preference values in a given Pareto-efficient state of affairs. For example, if individuals do not care about the values of equality and liberty as a minimal inviolable personal domain (as in Sen 1970a), then there can be a Pareto-efficient state which exhibits a win-win relationship between these non-preference values.

Proposition 4 (Conflict of value-efficiency and Pareto-efficiency) If $P \subset V$, then there may exist a value-efficient state of affairs that is not Pareto-efficient, and vice versa, a Pareto-efficient state that is not value-efficient.²²

Proof. See Appendix A.6

Generally, a Pareto-efficient state of affairs is not value-efficient if there exists another state of affairs where people are equally well off and some non-preference value is attained to a higher degree. For example, Dasgupta (Dasgupta 2005: 240) considers the case of the liberal paradox, where a state that respects liberty can be Pareto-inefficient, but is value-efficient as illustrated in Figure 3 through state of affairs y^b .

5 Conclusion

We have studied relationships between values on the set of feasible states of affairs by following a consequentialist definition of value that focuses on the assessment of states of affairs. We have shown that for value sets with three or more values, there can be independence, win-win relationships or trade-offs between values in value-efficient states of affairs. That is, neither win-win nor independence relationships between values indicate value-inefficient states of affairs, if there are more than two values. We have demonstrated that in any case the absence of trade-offs between values indicates a value-inefficient state of affairs. With regard to opportunity costs, it has become clear that trade-offs are associated with positive opportunity costs, win-win relationships

²² This is a comparatively weak statement. Kaplow and Shavell (2001) make a stronger statement yet they refer to the set of all conceivable states, and not explicitly to the set of feasible states. It is clear that a general statement regarding the conflict between value and Pareto-efficiency in the feasible set requires much more detailed assumptions regarding values and the feasible set as the broad literature on social choice theory shows.

with negative opportunity costs and independence relationships with negative or zero opportunity costs.

The practical relevance of these results to decision-makers is the following. First, and foremost, the value set must be complete and clear for decision-making, as value-efficiency and the indication of (in)efficiency, crucially hinges upon the values and the number of values in the value set. Second, a trade-off between *two* values indicates a state of affairs that is efficient with respect to *all* values. Such a trade-off can be recognized from positive opportunity costs of some value. Third, a win-win relationship between two values does not imply a need to change the state of affairs. For, giving in to a win-win relationship between two values may incur a trade-off between one of those values and a third value.

Regarding the connection between Pareto-efficiency and value-efficiency, we have shown that the former is a special case of the latter when individual preferences are taken to be the only values. If there are values that are not reducible to individual preferences, as in some ethical theories, we have seen that there can be win-win relationship between non-preference values in a Pareto-efficient state of affairs.

The practical relevance of these results to decision-makers is the following. If there is at least one non-preference value, Pareto-efficiency cannot taken to be an indication of value-efficiency. This matters in all practical contexts where Pareto-efficiency is taken as the criterion of a "good" state of affairs, for example when justifying institutions like competitive markets, free trade or anti-trust regulation. This underscores how important the conception of well-being is for the analysis of conflicts of non-preference values with Pareto-efficiency as this conception determines what constitutes individual preferences and, consequently, which values are preference-based and which are not (e.g. Sen 1980, Anderson 1993).

While these results are quite general, they are not derived for a concrete ethical problem that generates a specific set of feasible states of affairs. That is, in a detailed economic model results on relationships between particular values could be derived from more specific assumptions on individual preferences, instruments and resources. In this vein, the general insights into value-efficiency and relationships between values in this paper could be applied for concrete ethical theories such as Sen's capability approach (e.g. Pattanaik and Xu 2012) or the relationship between intra- and intergenerational justice (e.g. Baumgärtner et al. 2012).

Acknowledgements

We are grateful to Dave Abson, Urs Allenspacher, Georg Brun, John-Oliver Engler, Gertrude Hirsch-Hadorn, Ivo Wallimann-Helmer, Michael Liebendörfer, Dominic Roser, Michael Schefczyk, Fabian Schuppert, Martin Quaas, the team of the Chair of Economics / Resource Economics at ETH Zürich, for critical and constructive discussion, and to the German Federal Ministry of Education and Research for financial support under grant 01UN1011A.

References

- Alesina, A., Di Tella, R., and MacCulloch, R. (2004). Inequality and happiness: are Europeans and Americans different? *Journal of Public Economics*, 88:2009–2042.
- Anderson, E. (1993). Value in Ethics and Economics. Harvard University Press, Cambridge, Mass.
- Arrow, K. (1951). Social Choice and Individual Values. Wiley, New York.
- Barry, B. (1965). Political Argument. Routledge and Keagan Paul, London.
- Baland, J.M, Bardhan, P., Das, S., Mookherjee, D., and Sarkar, R. (2010). The environmental impact of poverty: Evidence from firewood collection in rural nepal. *Economic Development and Cultural Change*, 59(1):23–61.
- Baumgärtner, S., Glotzbach, S., Hoberg, N., Quaas, M.F., Stumpf, K. (2012). Economic analysis of trade-offs between justices. *Intergenerational Justice Review*, 1:4–9.
- Bergstrom, T. (1975). Maximal elements of acyclic relations on compact sets. *Journal* of *Economic Theory*, 10(3):403–404.

- Bentham, J. (1907). An Introduction to the Principles of Morals and Legislation. Clarendon Press, Oxford.
- Bentham, J. (1988). A Fragment on Government. Cambridge University Press, Cambridge.
- Berlin, I. (1969). Four Essays on Liberty. Oxford University Press, Oxford.
- Björklund, A., and Jäntti, M. (1997). Intergenerational income mobility in Sweden compared to the United States. *American Economic Review*, 87(5):1009–1018.
- Blank, M.R. (2002). Can equity and efficiency complement eachother? Labour Economics, 9(4):451–468.
- Boot, M. (2009). Parity, incomparability and rationally justified choice. *Philosophical Studies*, 146(1):75–92.
- Broome, J. (1991). Weighing Goods. Equality, Uncertainty and Time. Basil Blackwell Inc., Cambridge, Mass.
- Brun, B. and Tungodden, B. (2004). Non-welfaristic theories of justice: Is "the intersection approach" a solution to the indexing impasse? *Social Choice and Welfare*, 22(1):49–60.
- Brun, G. and Hirsch-Hadorn, G. (2008). Ranking policy options for sustainable development. *Poiesis and Praxis*, 5(1):15–31.
- Buchanan, J.M. (1959). Positive economics, welfare economics, and political economy. Journal of Law and Economics, 2:124–138.
- Chang, R.(ed.) (1997a). Incommensurability, Incomparability and Practical Reason.Harvard University Press, Cambridge, Mass.
- Chang, R. (1997b). Introduction. In: Chang, R. (ed.), Incommensurability, Incomparability and Practical Reason, Harvard University Press, Cambridge, Mass, pp. 1–34.

- Chan, R. (2002 a). The Possibility of Parity. *Ethics*, 112(4):659–688.
- Chan, R. (2002 b). Making Comparisons Count. Routledge, New York.
- Chang, R. (2005). Parity, interval value and choice. *Ethics*, 115(2):331–350.
- Copp, D. (2006). Introduction: Metaethics and normative ethics. In: Copp, D. (ed.), The Oxford Handbook of Ethical Theory, Oxford University Press, Oxford, pp. 357– 359.
- Corak, M. (2013). Income inequality, equality of opportunity, and intergenerational mobility. *Journal of Economic Perspectives*, 27(3):79–102.
- Cowen, T. (2007). The importance of defining the feasible set. *Economics and Philoso-phy*, 23(1):1–14.
- Dasgupta, P. (2005). What do economists analyzse and why: values or facts? *Economics and Philosophy*, 21(2):221–278.
- De Haan, J., Lundström, S., and Sturm, J.-E. (2006). Market-oriented institutions and policies and economic growth: a critical survey. *Journal of Economic Surveys*, 20(2):157–191.
- Engel, S., Pagiola, S., and Wunder, S. (2008). Designing payments for environmental services in theory and practice: An overview of the issues. *Ecological Economics*, 65(4):663–674.
- Fleurbaey, M., Tungodden, B., and Chang, H. (2003). Any non-welfarist method of policy assessment violates the Pareto principle: A comment. *Journal of Political Economy*, 111(6):613–639.
- Fleurbaey, M., Suzumura, K., and Tadenuma, K. (2005). The informational basis of the theory of fair allocation. Social Choice and Welfare, 24(2):311–341.
- Fleurbaey, M. (2007). Social choice and the indexing dilemma. Social Choice and Welfare, 29(4):633–648.

- Glotzbach, S., and Baumgärtner, S. (2012). The relationship between intragenerational and intergenerational ecological justice. *Environmental Values*, 21(3):331–335.
- Griffin, J. (1986). Well-Being: Its Meaning, Measurement and Moral Importance. Clarendon Press, Oxford.
- Hausman, D. (1993). Structure of good. A review of weighing goods by John Broome. Ethics, 103(4):792–806.
- Hausman, D. and McPherson, M. (1993). Taking ethics seriously: Economics and contemporary moral philosophy. *Journal of Economic Literature*, 31(2):671–731.
- Hougaard, J. and Tvede, M. (2001). The existence of maximal elements: generalized lexicographic relations. *Journal of Mathematical Economics*, 36(2):111–115.
- Houy, M. and Tadenuma, K. (2009). Lexicographic compositions of multiple criteria for decision making. *Journal of Economic Theory*, 144(4):1770–1782.
- Hsieh, N. (2008). Incommensurable values. In: Zalta, E. (ed.), The Stanford Encyclopedia of Philosophy, Fall 2008 Edition, [Online available at: http://plato.stanford. edu/archives/fall2008/entries/value-incommensurable/]
- Hurka, T. (2006). Value Theory. In: Copp, D. (ed.), The Oxford Handbook of Ethical Theory, Oxford University Press, Oxford, pp. 357–359.
- Kaplow, L. (2007). Primary goods, capabilities,... or well-being? The Philosophical Review, 116(4):603–632.
- Kaplow, L. and Shavell, S. (2001). Any non-welfarist method of policy assessment violates the Pareto principle. *Journal of Political Economy*, 109(2):281–286.
- Le Grand, J. (1990). Equity versus efficiency: the elusive trade-off. *Ethics*, 100(3):554–568.
- Mason, E. (2011). Value pluralism. In: Zalta, E. (ed.), The Stanford Encyclopedia of Philosophy, Fall 2011 Edition, [Online available at: http://plato.stanford.edu/ archives/fall2011/entries/value-pluralism/]

- MasColell, A., Whinston, M.D., and Green, J. (1995). *Microeconomic Theory*. Oxford University Press, Oxford.
- Nagel, T. (1979). The fragmentation of value. In: Nagel, T. (ed.), Mortal Questions, Cambridge University Press, Cambridge, pp. 128–141.
- Okun, A. (1975). Equality and Efficiency: the Big Trade-Off. Brookings Institution, Washington.
- Pattanaik, P. and Xu, Y. (2012). On dominance and context-dependence in decisions involving multiple attributes. *Economics and Philosophy*, 28(2):117-132.
- Putterman, L. Roemer, J., and Silvestre, J. (1998). Does egalitarianism have a future? Journal of Economic Literature, 36(2):861-902.
- Rabinowicz, W. (2008). Value relations. *Theoria*, 74(1):18–49.
- Rawls, J. (1971). A Theory of Justice. Harvard University Press, Cambridge, Mass.
- Rawls, J. (1987). Basic liberties and their priority. In: McMurrin, S. (ed.), Liberty, equality and law: Selected Tanner lectures on moral philosophy, Cambridge University Press, Cambridge, pp. 1–87.
- Raz, J. (1997). Incommensurability and agency. In: Chang, R. (ed.), Incommensurability, Incomparability and Practical Reason, Harvard University Press, Cambridge, Mass, pp. 1–34.
- Raz, J. (1999). Engaging Reason: On the Theory of Value and Action. Oxford University Press, Oxford.
- Roemer, J. (1996). Theories of Distributive Justice. Harvard University Press, Cambridge, Mass.
- Scanlon, T. (1998). What We Owe Eachother. Belknap Press of Harvard University Press, Cambridge, Mass.

- Sen, A. (1970a). The impossibility of a paretian liberal. Journal of Political Economy, 78(1):152–157.
- Sen, A. (1970b). Collective Choice and Social Welfare. Holden-Day, San Francisco.
- Sen, A. (1977). Rational fools: a critique of the behavioral foundations of economic theory. *Philosophy and Public Affairs*, 6(4):317–344.
- Sen, A. (1979a). Personal utilities and public judgements: or what's wrong with welfare economics. *Economic Journal*, 89:537–558.
- Sen, A. (1979b). Utilitarianism and welfarism. Journal of Philosophy, 76(9):463–489.
- Sen, A. (1980). Equality of what? In: McMurrin, S. (ed), The Tanner Lecture on Human Values, Vol. 1, Cambridge University Press, Cambridge, pp. 197-220.
- Sen, A. (1985). Well-Being, agency and freedom: the Dewey Lectures 1984. Journal of Philosophy, 82(4):169–221.
- Sen, A. (1997). Maximization and the act of choice. *Econometrica*, 65(4):745–779.
- Sen, A. (1999). The possibility of social choice. American Economic Review, 89(3):349– 378.
- Sen, A. (2000). Consequential evaluation and practical reason. *Journal of Philosophy*, 97(9):477–502.
- Stern, N. (2007). The Economics of Climate Change: The Stern Review. Cambridge University Press, Cambridge.
- Sinnott-Armstrong, W. (2012). Consequentialism. In: Zalta, E. (ed.), The Stanford Encyclopedia of Philosophy, Winter 2012 Edition, [Online available at: http:// plato.stanford.edu/archives/win2012/entries/consequentialism/]
- Suzumura, K. (1987). Social welfare function. In: Eatwell, J., Milgate, M. and Newman, P. (eds), *The New Palgrave: A Dictionary of Economics*, Vol. 4, Macmillan, London, pp. 418–420.

- Walker, M. (1977). On the existence of maximal elements. Journal of Economic Theory, 16:470–474.
- Williams, B. (1981). Moral Luck: Philosophical Papers. Cambridge University Press, Cambridge.
- World Bank (2012). Inclusive Green Growth: The Pathway to Sustainable Development.World Bank, Washington, D.C.
- Zimmerman, M. (2010). Intrinsic vs. extrinsic Value. In: Zalta, E. (ed.), The Stanford Encyclopedia of Philosophy, Winter 2010 Edition, [Online available at: http://plato.stanford.edu/archives/win2010/entries/value-intrinsic-extrinsic/]

Appendix

A.1 Proof of Proposition 1

(i): Consider two value-efficient states of affairs $y, y' \in Y$ and a value set with #V = lvalues. For two values #V = l = 2 and $v, w \in V$, value-efficiency of y, y' requires a trade-off between v and w, i.e. $y \prec_v y', y \succ_w y'$. For more than two values #V = l > 2, value-efficiency of y, y' requires a trade-off between at least two values. To show this assume there exists no trade-off between any two values in y, i.e. $y \prec_v y'$ or $y \sim_v y'$ for all $v \in V$, then y' is not value-efficient as Definition 7 that there must exists no other state of affairs $y' \in Y$ for which $y' \succeq_v y$ for all $v \in V$ and $y' \succ_v y$ for at least one $v \in V$.

(ii): Take two values v, w from a value set V with #V = l values and any two states of affairs $y, y' \in Y$. If $y \prec_v y'$ does necessarily lead to $y \succ_w y'$ for some $w \in V$, then y is a value-efficient state of affairs as Definition 7 says there must exist no $y' \in Y$ for which $y \prec_v y'$ for some $v \in V$ and $y \preceq_w y'$ for all $w \in V$.

(iii): Assume there are 3 values in the value set $u, v, w \in V$ and two states of affairs $y, y' \in Y$. Assume the orderings necessary for a win-win relationship between u and v: $y \prec_u y', y \prec_v y'$; and show that $y, y' \in Y$ can still be value-efficient. Assume $y \succ_w y'$ so

that y' does not dominate y. This yields two value-efficient states of affairs y, y' where y exhibits a win-win relationship between $u, v \in V$.

To show an an independence relationship in a value-efficient state of affairs assume one further state of affairs y''. Assume additionally the orderings necessary for an independence relationship between u and $v: y \prec_u y'', y \sim_v y''$; and show that $y, y', y'' \in$ Y can still all be value-efficient. Assume $y \succ_w y', y \succ_w y''$ so that y' and y'' do not dominate y. Assume $y' \sim_w y''$ and $y' \prec_u y'', y' \succ_v y''$ so that y' does not dominate y''. This yields three value-efficient states of affairs y, y', y'' where y exhibits an independence relationship between $u, v \in V$.

A.2 Proof of Proposition 2

Follows immediately from Definitions 8 and 9.

A.3 Proof of Corollary 1

Follows immediately from Propositions 1 and 2.

A.4 Proof of Proposition 3

If all $v \in V$ are individual preferences, then all value relations are identical to preference relations. This makes value-efficiency equivalent to Pareto-efficiency.

A.5 Proof of Corollary 2

Trade-off relationship

Assume there are 4 values in the value set: two individual preferences $p_1, p_2 \in P \subset V$ and two non-preference values $v, w \in V$ and two states of affairs y, y'. Assume the orderings necessary for both states to be Pareto-efficient: for the first individual $y \prec_{p_1} y'$, and for the second individual $y \succ_{p_2} y'$. Further, assume orderings for a trade-off between v and $w: y \prec_v y'$ and $y \succ_w y'$. This yields two Pareto-efficient and value-efficient states of affairs y, y' where y exhibits a trade-off relationship between $v, w \in V$.

Win-win relationship

Assume there are 4 values in the value set: two individual preferences $1, 2 \in P \subset V$ and two non-preference values $v, w \in V$ and two states of affairs y, y'. Assume the orderings necessary for both states to be Pareto-efficient: for the first individual $y \prec_{p_1} y'$, and for the second individual $y \succ_{p_2} y'$. Further, assume orderings for a win-win relationship between v and w: $y \prec_v y'$ and $y \prec_w y'$. This yields two Pareto-efficient and valueefficient states of affairs y, y' where y exhibits a trade-off relationship between $v, w \in V$.

Independence relationship

Assume there are 4 values in the value set: two individual preferences $1, 2 \in P \subset V$ and two non-preference values $v, w \in V$ and three states of affairs y, y', y''. Assume the orderings necessary for all three states to be Pareto-efficient: for the first individual $y \prec_{p_1} y', y \prec_{p_1} y'', y' \prec_{p_1} y''$, and for the second individual $y \succ_{p_2} y', y \succ_{p_2} y'', y' \succ_{p_2} y''$. Further, assume orderings for an independence relationship between v and w: $y \prec_v y'$, $y \prec_v y''$ and $y \prec_w y', y \sim_w y''$. Assume $y' \sim_v y''$ and $y' \prec_w y''$, so that y' does not dominate y''. This yields three Pareto-efficient and value-efficient states of affairs y, y', y''where y exhibits an independence relationship between $v, w \in V$.

A.6 Proof of Proposition 4

Assume a value set with three values: two individual preferences $p_1, p_2 \in P \subset V$ and a non-preference value $v \in V$ and two states of affairs y, y'. Then it can be that $y \prec_{p_1} y'$ and $y \prec_{p_2} y'$, but $y \succ_v y'$. This yields two value-efficient states of affairs y, y', where state y is not Pareto-efficient.

Working Paper Series in Economics

(recent issues)

- No.310: *Joachim Wagner:* Exports, R&D and Productivity: A test of the Bustos-model with enterprise data from France, Italy and Spain, October 2014
- No.309: *Thomas Wein:* Preventing Margin Squeeze: An Unsolvable Puzzle for Competition Policy? The Case of the German Gasoline Market, September 2014
- No.308: *Joachim Wagner:* Firm age and the margins of international trade: Comparable evidence from five European countries, September 2014
- No.307: John P. Weche Gelübcke: Auslandskontrollierte Industrie- und Dienstleistungsunternehmen in Niedersachsen: Performancedifferentiale und Dynamik in Krisenzeiten, August 2014
- No.306: *Joachim Wagner:* New Data from Official Statistics for Imports and Exports of Goods by German Enterprises, August 2014
- No.305: *Joachim Wagner:* A note on firm age and the margins of imports: First evidence from Germany, August 2014
- No.304: Jessica Ingenillem, Joachim Merz and Stefan Baumgärtner: Determinants and interactions of sustainability and risk management of commercial cattle farmers in Namibia, July 2014
- No.303: Joachim Wagner: A note on firm age and the margins of exports: First evidence from Germany, July 2014
- No.302: *Joachim Wagner:* A note on quality of a firm's exports and distance to destination countries: First evidence from Germany, July 2014
- No.301: Ahmed Fayez Abdelgouad: Determinants of Using Fixed-term Contracts in the Egyptian Labor Market: Empirical Evidence from Manufacturing Firms Using World Bank Firm-Level Data for Egypt, July 2014
- No.300: Annika Pape: Liability Rule Failures? Evidence from German Court Decisions, May 2014
- No.299: Annika Pape: Law versus Economics? How should insurance intermediaries influence the insurance demand decision, June 2013
- No.298: Joachim Wagner: Extensive Margins of Imports and Profitability: First Evidence for Manufacturing Enterprises in Germany, May 2014 [published in: Economics Bulletin 34 (2014), 3, 1669-1678]
- No.297: *Joachim Wagner:* Is Export Diversification good for Profitability? First Evidence for Manufacturing Enterprises in Germany, March 2014 [published in: Applied Economics 46 (2014), 33, 4083-4090]
- No.296: *Joachim Wagner:* Exports and Firm Profitability: Quality matters!, March 2014 [published in: Economics Bulletin 34 (2014), 3, 1644-1652]
- No.295: *Joachim Wagner:* What makes a high-quality exporter? Evidence from Germany, March 2014 [published in: Economics Bulletin 34 (2014), 2, 865-874]
- No.294: Joachim Wagner: Credit constraints and margins of import: First evidence for German manufacturing enterprises, February 2014

- No.293: *Dirk Oberschachtsiek*: Waiting to start a business venture. Empirical evidence on the determinants., February 2014
- No.292: *Joachim Wagner*: Low-productive exporters are high-quality exporters. Evidence from Germany, February 2014 [published in: Economics Bulletin 34 (2014), 2, 745-756]
- No.291: Institut für Volkswirtschaftslehre: Forschungsbericht 2013, Januar 2014
- No.290: Stefan Baumgärtner, Moritz A. Drupp und Martin F. Quaas: Subsistence and substitutability in consumer preferences, December 2013
- No.289: *Dirk Oberschachtsiek*: Human Capital Diversity and Entrepreneurship. Results from the regional individual skill dispersion nexus on self-employment activity., December 2013
- No.288: Joachim Wagner and John P. Weche Gelübcke: Risk or Resilience? The Role of Trade Integration and Foreign Ownership for the Survival of German Enterprises during the Crisis 2008-2010, December 2013
- No.287: Joachim Wagner: Credit constraints and exports: A survey of empirical studies using firm level data, December 2013
- No.286: *Toufic M. El Masri*: Competition through Cooperation? The Case of the German Postal Market, October 2013
- No.285: *Toufic M. El Masri*: Are New German Postal Providers Successful? Empirical Evidence Based on Unique Survey Data, October 2013
- No.284: Andree Ehlert, Dirk Oberschachtsiek, and Stefan Prawda: Cost Containment and Managed Care: Evidence from German Macro Data, October 2013
- No.283: Joachim Wagner and John P. Weche Gelübcke: Credit Constraints, Foreign Ownership, and Foreign Takeovers in Germany, September 2013
- No.282: *Joachim Wagner*: Extensive margins of imports in The Great Import Recovery in Germany, 2009/2010, September 2013 [published in: Economics Bulletin 33 (2013), 4, 2732-2743]
- No.281: Stefan Baumgärtner, Alexandra M. Klein, Denise Thiel, and Klara Winkler: Ramsey discounting of ecosystem services, August 2013
- No.280: Antonia Arsova and Deniz Dilan Karamen Örsal: Likelihood-based panel cointegration test in the presence of a linear time trend and cross-sectional dependence, August 2013
- No.279: *Thomas Huth*: Georg von Charasoff's Theory of Value, Capital and Prices of Production, June 2013
- No.278: Yama Temouri and Joachim Wagner: Do outliers and unobserved heterogeneity explain the exporter productivity premium? Evidence from France, Germany and the United Kingdom, June 2013 [published in: Economics Bulletin, 33 (2013), 3, 1931-1940]
- No.277: *Horst Raff and Joachim Wagner*: Foreign Ownership and the Extensive Margins of Exports: Evidence for Manufacturing Enterprises in Germany, June 2013
- No.276: Stephan Humpert: Gender Differences in Life Satisfaction and Social Participation, May 2013
- No.275: Sören Enkelmann and Markus Leibrecht: Political Expenditure Cycles and Election Outcomes Evidence from Disaggregation of Public Expenditures by Economic Functions, May 2013

- No.274: Sören Enkelmann: Government Popularity and the Economy First Evidence from German Micro Data, May 2013
- No.273: *Michael Berlemann, Soeren Enkelmann,* and *Torben Kuhlenkasper*: Unraveling the Relationship between Presidential Approval and the Economy A Multi-Dimensional Semi-Parametric Approach, May 2013
- No.272: *Michael Berlemann* and *Sören Enkelmann*: The Economic Determinants of U.S. Presidential Approval A Survey, May 2013
- No.271: Soeren Enkelmann: Obama and the Macroeconomy Estimating Social Preferences Between Unemployment and Inflation, May 2013
- No.270: Anja Köbrich León: Does Cultural Heritage affect Employment decisions Empirical Evidence for Second Generation Immigrants in Germany, April 2013
- No.269: Anja Köbrich León and Christian Pfeifer: Religious Activity, Risk Taking Preferences, and Financial Bahavior, April 2013
- No.268: Anja Köbrich León: Religion and Economic Outcomes Household Savings Bahavior in the USA, April 2013
- No.267: John P. Weche Gelübcke and Isabella Wedl: Environmental Protection of Foreign Firms in Germany: Does the country of origin matter?, April 2013
- No.266: Joachim Wagner: The Role of extensive margins of exports in The Great Export Recovery in Germany, 2009/2010, March 2013 [published in: Jahrbücher für Nationalökonomie und Statistik 234 (2014), 4, 519-526]
- No.265: *John-Oliver Engler* and *Stefan Baumgärtner*: Model choice and size distribution: a Bayequentist approach, February 2013
- No.264: *Chiara Franco* and *John P. Weche Gelübcke*: The death of German firms: What role for foreign direct investment?, February 2013
- No.263: *Joachim Wagner*: Are low-productive exporters marginal exporters? Evidence from Germany, February 2013 [published in Economics Bulletin 33 (2013), 1, 467-481]
- No.262: Sanne Hiller, Philipp J. H. Schröder, and Allan Sørensen: Export market exit and firm survival: theory and first evidence, January 2013
- No.261: Institut für Volkswirtschaftslehre: Forschungsbericht 2012, Januar 2013
- No.260: Alexander Vogel and Joachim Wagner: The Impact of R&D Activities on Exports of German Business Services Enterprises : First Evidence from a continuous treatment approach, December 2012
- No.259: *Christian Pfeifer*: Base Salaries, Bonus Payments, and Work Absence among Managers in a German Company, December 2012
- No.258: Daniel Fackler, Claus Schnabel, and Joachim Wagner: Lingering illness or sudden death? Pre-exit employment developments in German establishments, December 2012 [published in: Industrial and Corporate Change 23 (2014), 4, 1121-1140]
- No.257: Horst Raff and Joachim Wagner: Productivity and the Product Scope of Multi-product Firms: A Test of Feenstra-Ma, December 2012 [published in: Economics Bulletin, 33 (2013), 1, 415-419]

(see www.leuphana.de/institute/ivwl/publikationen/working-papers.html for a complete list)

Leuphana Universität Lüneburg Institut für Volkswirtschaftslehre Postfach 2440 D-21314 Lüneburg Tel.: ++49 4131 677 2321 email: brodt@leuphana.de

www.leuphana.de/institute/ivwl/publikationen/working-papers.html